Modeling Within-Host Effects of Drugs on *Plasmodium falciparum* Transmission and Prospects for Malaria Elimination

Geoffrey L. Johnston^{1,2,3}, Peter W. Gething⁴, Simon I. Hay⁴, David L. Smith^{3*}, David A. Fidock^{1,5*}

 Department of Microbiology and Immunology, Columbia University College of Physicians and Surgeons, New York, New York, United States of America, 2 School of International and Public Affairs, Columbia University, New York, New York, United States of America, 3 Bloomberg School of Public Health, John Hopkins University, Baltimore, Maryland, United States of America, 4 Spatial Ecology and Epidemiology Group, Department of Zoology, University of Oxford, Oxford, United Kingdom,
Division of Infectious Diseases, Department of Medicine, Columbia University College of Physicians and Surgeons, New York, New York, United States of America

Abstract

Achieving a theoretical foundation for malaria elimination will require a detailed understanding of the guantitative relationships between patient treatment-seeking behavior, treatment coverage, and the effects of curative therapies that also block Plasmodium parasite transmission to mosquito vectors. Here, we report a mechanistic, within-host mathematical model that uses pharmacokinetic (PK) and pharmacodynamic (PD) data to simulate the effects of artemisinin-based combination therapies (ACTs) on Plasmodium falciparum transmission. To contextualize this model, we created a set of global maps of the fold reductions that would be necessary to reduce the malaria $R_{\rm C}$ (i.e. its basic reproductive number under control) to below 1 and thus interrupt transmission. This modeling was applied to low-transmission settings, defined as having a $R_0 < 10$ based on 2010 data. Our modeling predicts that treating 93–98% of symptomatic infections with an ACT within five days of fever onset would interrupt malaria transmission for ~91% of the at-risk population of Southeast Asia and ~74% of the global at-risk population, and lead these populations towards malaria elimination. This level of treatment coverage corresponds to an estimated 81-85% of all infected individuals in these settings. At this coverage level with ACTs, the addition of the gametocytocidal agent primaguine affords no major gains in transmission reduction. Indeed, we estimate that it would require switching \sim 180 people from ACTs to ACTs plus primaguine to achieve the same transmission reduction as switching a single individual from untreated to treated with ACTs. Our model thus predicts that the addition of gametocytocidal drugs to treatment regimens provides very small population-wide benefits and that the focus of control efforts in Southeast Asia should be on increasing prompt ACT coverage. Prospects for elimination in much of Sub-Saharan Africa appear far less favorable currently, due to high rates of infection and less frequent and less rapid treatment.

Citation: Johnston GL, Gething PW, Hay SI, Smith DL, Fidock DA (2014) Modeling Within-Host Effects of Drugs on *Plasmodium falciparum* Transmission and Prospects for Malaria Elimination. PLoS Comput Biol 10(1): e1003434. doi:10.1371/journal.pcbi.1003434

Editor: Neil M. Ferguson, Imperial College London, United Kingdom

Received July 3, 2013; Accepted November 25, 2013; Published January 23, 2014

Copyright: © 2014 Johnston et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: DLS, PWG and DAF are supported by separate grants from the Bill and Melinda Gates Foundation (OPP49446, OPP1068048 and OPP1040399 respectively). PWG is a Medical Research Council Career Development Fellow (#K00669X). SIH is funded by a Senior Research Fellowship from the Wellcome Trust (#095066). GLJ received funding from the National Science Foundation Graduate Research Fellowship Program. DLS and SIH also acknowledge support from the RAPIDD program of the Science & Technology Directorate, Department of Homeland Security, and the Fogarty International Center, National Institutes of Health. Authors also received funding from the Bloomberg Family Foundation (DLS, GL). Partial funding for this work was provided by the US National Institutes of Health (grant R01 Al079709, to DAF). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* E-mail: dlsmith@jhsph.edu (DLS); df2260@columbia.edu (DAF)

Introduction

Plasmodium falciparum, the most virulent of the *Plasmodium* species that cause malaria in humans, is responsible for hundreds of millions of cases per year [1]. The number of fatal outcomes is a matter of considerable debate, with estimates for 2010 ranging from 655,000 to 1,238,000 [2,3]. Studies nonetheless agree that overall levels of morbidity and mortality have declined over the past decade, due at least in part to the worldwide scaling up of insecticide-treated bed nets and the use of artemisinin-based combination therapies (ACTs). ACTs, which pair fast-acting short-lived artemisinin derivatives with longer-lasting partner drugs, are now the first-line antimalarial drugs in almost the entire malaria-endemic world [4,5].

Public health and malaria infection experts are increasingly promoting the goal of malaria elimination in areas of low transmission [6,7] while planning ways to achieve significant reductions in higher-transmission areas [8,9]. Major obstacles, however, stand in the way. These include insecticide and drug resistance [10,11], under-developed health care systems, shifting public funding priorities, donor fatigue [6], malaria importation [12] and economic constraints [13]. The complex life cycle of *P. falciparum* also presents unique challenges [14]. *P. falciparum* stages differ markedly in their levels of metabolic activity, within-host locations, and susceptibilities to antimalarials. Mathematical modeling can help guide elimination efforts by providing quantitative predictions to assess the feasibility of different intervention and control strategies [15–18].

Author Summary

We utilize a within-host mathematical model of malaria transmission to predict the effects of antimalarial treatment across the globe. We predict that areas containing 91% of the at-risk population of Southeast Asia can achieve elimination if at least 93–98% of symptomatic individuals are promptly treated with effective artemisinin-based combination therapies (ACTs), based on assessments of treatment and transmission levels as of 2010. The benefit of attaining this level of coverage far outperforms that of adding additional gametocyte-specific transmission-blocking drugs to current ACTs. We advocate for elimination programs in Southeast Asia to focus on maximizing ACT coverage.

ACTs and other antimalarial drugs reduce transmission in three ways: by killing the disease-causing asexual blood stages and thus preventing continued production of the intra-erythrocytic sexual gametocyte forms; by killing existing gametocytes and reducing or preventing onward transmission to the mosquito (thereby reducing parasite oocyst numbers in mosquito midguts); and by post-treatment drug prophylaxis wherein residual drug levels can protect against new infections [19]. Here we utilize mathematical modeling to quantify how ACTs reduce malaria transmission, with or without late-stage gametocytocidal agents such as primaquine (PQ) or methylene blue. These agents are receiving considerable interest within the malaria community as to how their action might be leveraged to help interrupt parasite transmission [20–22].

Here, we report outputs from our within-host model of *P. falciparum* infection and transmission, and overlay these findings onto geospatial maps of malaria endemicity in order to predict the benefits of extended coverage of infected individuals and incorporation of transmission-blocking agents into current ACT regimens. In low-transmission settings we predict that if at least 93–98% of all symptomatic infections, corresponding to an estimated 81–85% of all infected individuals, were treated within five days of first fever, then the $R_{\rm C}$ could be reduced to below one and malaria would progress towards elimination in regions harboring over ~91% of at-risk populations in Southeast Asia and ~74% of the global at-risk population. Our findings suggest that increasing treatment coverage with ACTs would be more effective than adding additional transmission-blocking agents in driving towards the goal of malaria elimination in Southeast Asia.

Results

A within-host model to predict the effects of drug treatment on reducing malaria transmission

We used our recently developed within-host model of the progression of *P. falciparum* infection [23] to simulate the densities of asexual blood stage parasites and gametocytes in a population of individuals with no acquired immunity to malaria. The variability in densities among individuals was matched to the variability observed in malaria therapy studies, in which syphilitic individuals with no history of malaria infection were infected with *P. falciparum* to induce a fever and clear the syphilis infection [23]. Of note, our model incorporates three different types of antimalarial immunity: an innate response that establishes an upper limit for parasite density; a PfEMP1 variant-specific response that regulates short-term periodic oscillations in density; and a variant-transcending response that causes a steady log-linear decrease in density over time, clearing the infection [23]. In our simulations, these responses were calibrated such that the infection dynamics

matched those of experimental challenge volunteers who had no previous malarial infections, i.e., we assumed that individuals either had no prior episode of malaria or had acquired malaria so long ago that their responses were equivalent to that of individuals without prior infections. Further, we simulated only single infections, i.e. we did not simulate infections that overlapped in time.

Figure 1 illustrates six runs from our within-host infection model; untreated individuals are denoted by 'Untreated' or 'U'. **Figure 1A** shows the \log_{10} parasitized red blood cells (PRBC) per µL, while **Figure 1B** depicts the daily gametocytemias over time, which are typically two logs lower for untreated individuals. The **Figure 1A** inset illustrates the asexual densities for the first 50 days post emergence of parasites into the bloodstream; colored triangles illustrate the onset of first fever [23]. Once the daily gametocytemias were simulated, a gametocyte density-to-infectivity relationship was utilized to translate these values into predicted infectiousness to mosquitoes over time. **Figure 1C** illustrates the predicted human-to-mosquito infectivity for each of the three untreated individuals (U) using the Jeffery-Eyles (JE) relationship between gametocyte densities and infectivity [23].

As described in detail below, we then incorporated the effects of drug treatment into our within-host model to illustrate how this modeled treatment affects malaria transmission. Figure 1 includes the results of a hypothetical drug treatment on three individuals infected with P. falciparum; treated individuals are denoted by 'Treated' or 'T'. As with the untreated cases, Figure 1A shows the asexual parasite densities in log₁₀ PRBC/ µL, while Figs. 1B and 1C depict the daily gametocytemias and human-to-mosquito infectivities. Treatment was assumed to begin 4, 9 and 14 days after the onset of fever. The effects of drug treatment can be seen immediately on the asexual blood stage population, which showed a steep drop in numbers after dosing, as well as the gametocytemias that were lower among the treated individuals following treatment. To calculate the net infectivity of individuals to mosquitoes, gametocytes densities were transformed into infectivity probabilities, and the area under the infectivity curve was derived (AUIC). This approach was previously used with field data [24] to estimate onward infectivity following treatment. In the case where drugs reduced oocyst numbers independently of their effects against gametocytes, the gametocyte density-to-infectivity relationships were adjusted in a drugdependent manner. Infectivities of treated patients (T) were of relatively brief duration and had largely disappeared within 30 days (Figure 1C). This hypothetical example illustrates many of the processes involved in modeling the effects of drugs on transmission. Below we describe how we have used in vitro and field data to parameterize the various components of drug activity and predict the effects of various antimalarial therapies in realworld settings.

Modeling drug effects against asexual blood stage *P*. *falciparum* parasites

To model the effects of different drugs on asexual parasite densities, we first modeled the within-host concentrations of the partner drugs of two ACTs, artesunate+mefloquine (AM) and artemether+lumefantrine (AL). AM is a frequently used first-line therapy in parts of Southeast Asia [24], while AL has recently become the most widely-used ACT worldwide [25]. Because we calculated our asexual parasite densities daily, we did not model the explicit concentrations of the artemisinin derivatives, as these drugs have half-lives of 1 to 3 hr [4]. However, we did incorporate their fast-acting PD potency on asexual blood stage parasite densities [4]. As a point of reference, we also modeled the drug



Figure 1. Illustration of asexual, gametocyte, and human-tomosquito infectivity model outputs. The P. falciparum infection model was run six times to simulate three untreated individuals and another three treated with a hypothetical antimalarial. (A) Individual log₁₀ asexual blood stage parasitemias as a function of the number of days post emergence of parasites from the liver into the bloodstream. The inset depicts the first 50 days of infection; the triangles above indicate the first day of fever. In this example, three individuals were assumed to seek treatment a variable number of days after the onset of fever. The black line illustrates the approximate level of detectability by microscopy (~10 parasitized red blood cells/µL). (B) Daily gametocytemias (sexual stage parasitemias) of the same six individuals. The hypothetical drug treatment was assumed to target early stages of gametocyte development more strongly than later stages. (C) Estimated probability of human-to-mosquito parasite transmission for treated (T) vs. untreated (U) individuals. Areas under the infectivity curves (AUIC) are equivalent to the number of fully infectious days. Net infectivity simulations yielded 1.0, 2.4, and 6.3 fully infectious days for

treated (T) and 67.7, 6.5, and 28.6 days for untreated (U) individuals, respectively. The model outputs for untreated patients shown in panels A–C were previously reported in [23] and are shown to compare with our modeling of treated patients. doi:10.1371/journal.pcbi.1003434.g001

concentrations of chloroquine (CQ), the former first line therapy that is highly active against asexual blood stages and has some activity against very early stage gametocytes but is inactive against mature gametocytes [21]. For lumefantrine (LMF) uptake and clearance we used a two-compartment pharmacokinetic model parameterized from field data; for mefloquine (MFQ) and CQ we used two-compartment non-parametric models, also parameterized from field data. Model equations, model parameters, and a description of the model fitting to average concentrations as well as population variation are detailed in **Text S1**. Notably, our study assumed that individuals were fully compliant with treatment and that parasites were sensitive to the various drug combinations. In future work we hope to examine how differences in patient compliance and parasite drug susceptibility impact transmission.

Figure 2 (panels A, C, E) illustrates the results of our PK simulations for LMF, MFQ, and CQ, respectively. The black lines indicate the median (LMF) or mean (MFQ, CQ) of the population concentrations, while the blue lines indicate simulated individual concentration profiles. Model outputs revealed wide variations in concentrations within a population, due to differences in rates of drug uptake and clearance. The number of concentrations depicted in each panel corresponds to the number of patients in each of the studies that provided data for model fitting. Of note, LMF plasma concentrations achieved considerably higher levels than the other two agents.

To simulate the effects of these drug concentrations against asexual parasites, we first calculated hourly plasma and/or blood concentration levels from our PK modeling. We then translated the hourly drug concentrations into asexual activities, assuming that the dose-response relationships could be modeled as Hill functions. These Hill functions were parameterized from in vitro and field data (see the SI). The asexual activities of drugs were quantified as 48-hour parasite reduction ratios (PRRs), i.e., the fold decreases in parasite numbers every 48 hr, corresponding to one cycle of intra-erythrocytic development and reinvasion. Figure 2 (panels B, D, F) illustrates the drug concentrations translated into these PRRs over time. To calculate the asexual parasite densities in our model under the effects of drugs for a generic day t, we took the densities from day t-1, applied our within-host model to calculate densities on day t including both the effects of parasite growth and host immune responses, then multiplied densities by the square root of the mean 48-hour PRR for drug concentrations during day t. The resulting density was then used to calculate densities at *t*+1, and so on until the end of the simulation.

Regardless of the drug regimens simulated here, asexual parasite densities fell rapidly when an individual was treated. The PRRs nevertheless showed substantial differences between drugs in later time periods following treatment. For example, the LMF PPRs rapidly declined within 5–15 days of treatment, whereas CQ took longer to decline while showing more heterogeneity. MFQ was also heterogeneous but always showed lower PPRs even at peak plasma concentrations. The maximal PRRs, and the differential rates of absorption, clearance, and volumes of drug distribution are described more fully in the **SI**.

Predicted effects of antimalarials on gametocyte development and transmission

While effective drug treatment rapidly clears asexual blood stage parasites, even successful regimens differ markedly in their



Figure 2. Modeled pharmacokinetic and pharmacodynamic profiles of lumefantrine (LMF), mefloquine (MFQ) and chloroquine (CQ) against asexual blood stage parasites. (A) Modeled plasma concentrations of LMF. The wide variability in concentrations reflects individual differences in absorption and clearance. The black line indicates population median drug concentrations. (B) Estimated parasite reduction ratios (PRR) over a 48 hr blood stage cycle for LMF as a function of time post onset of treatment. PRR is the fold reduction in asexual parasite densities due to drug action. PRR values were calculated from drug concentrations using a Hill function transformation. Each curve illustrates the PRR over time for a simulated individual; the black line illustrates the PRR for population median drug concentrations. (**C–D**) Plasma concentrations for MFQ and corresponding activities against asexual blood stage parasites. Drug concentrations reflect data from both plasma and whole blood studies. (**E–F**) Modeled plasma concentrations are in µg/ml. doi:10.1371/journal.pcbi.1003434.g002

effects on gametocytes. As gametocytes develop, they become metabolically more inert, thus reducing the number of drug targets available. Specifically, gametocytes mature over the course of $\sim 10-15$ days through Stages I–V, with each stage differing in metabolic activity and drug susceptibility [26]. Stages I–IV are sequestered in the microvasculature, spleen or bone marrow and are not present in the blood circulation. Only Stage V gametocytes have matured to the point of releasing from sequestered sites, thus becoming visible by blood smear. Here, we simulated differential effects of drugs against each developmental stage. This feature is novel to the literature so far as we know, although a recent article did allow for a differential effect on noncirculating vs. circulating gametocytes [27].

To parameterize the modeled effects of drugs on gametocytes, we first conducted a literature review to examine the types of datasets that could inform the model. There are few *in vitro* studies that have measured the stage-specific effects of drugs on gametocytes [21]; however, many field studies have examined the clearance of gametocytes within a population after drug treatment. These studies typically measured the proportion of individuals who were gametocytemic by microscopy after treatment (threshold for detection: \sim 5–10 gametocytes per µL [28,29]).

Figure S1 illustrates the prevalence of gametocytes from field studies of patients treated with a variety of antimalarials [28,30–38]. The field studies are disaggregated by drug type: treatment with SP [28,30,33]; CQ or amodiaquine (AQ) (sometimes in combination with SP) [28,30–33]; various ACTs [28,30–32,34–36,38]; or ACTs plus PQ [35,36,38]. The inter-study variability observed in **Figure S1** was likely due to a variety of factors, including the levels of acquired immunity in the population, exact timing of treatment, age of treated individuals, differences in parasite biology, and drug treatment regimens.

These data were used to calibrate the gametocyte component of our drug effects model. To generate a set of model outputs to compare against field data, we simulated treating individuals with a three-day course of AM and varied the assumed killing properties of the two component drugs. We assumed that treatment started relatively early in the infection, i.e., 5 days after first fever, in agreement with field studies from Thailand and Indonesia [34,38]. Treatment timing will vary from place to place given the treatmentseeking behaviors of the local population; if treatment is delayed significantly beyond this point, the ability of drugs to reduce transmission is likely to be diminished. We began our simulations assuming that the treatment had no effect on gametocytes at all; we called this therapy purely schizonticidal. We then increased the simulated killing properties of the combination, assuming that the components only killed early stages of gametocytes (e.g. as for CQ). We gradually increased this presumed killing power against early gametocytes (from mild to varying levels of moderate to strong, as in **Figure 3A**): as the killing power increased, gametocyte prevalence post-treatment decreased. We then assumed that the short-lived component killed both early and late stage gametocytes, with a larger effect on the former (e.g. as for an ACT). Finally, we simulated adding a single dose of a third drug that killed both early and late stage gametocytes, with varying levels of activity (e.g. as for ACT+PQ). Additionally, we varied the timing of this third component as this has been a topic of debate [39], by assuming that treatment occurred either on the first day of ACT treatment, denoted by '<day 0>', or last day, denoted by '<day 2>'. Our studies used the simulated prevalence of post-treatment gametocyte positive individuals as the output to be compared against field data, since this metric was most often tracked in the field.

Once the model outputs had been generated, these were compared to field studies of drug treatments with similar activity (see Figures S1, S2): the modeled schizonticidal treatment data were compared to field trials with SP; mild to moderate gametocytocidal outputs were compared to data from field trials of CQ, CQ+SP or AQ+SP; strong gametocytocidal outputs were compared to ACT clinical trials data, and the modeled triplecombination data were compared to field data for ACTs+PQ. For each drug activity type we then chose the sets of simulations that most closely resembled the mean, maximum, and minimum of observed responses to represent the effects of each class of drugs against gametocytes. We also included some intermediate sets of simulations for the sake of comparison. Figure S2 illustrates the model outputs that best approximated the mean and observed variation in the field data; all model means were from 1,000 runs for each parameterization. We modeled the entire range of observed variation in post-treatment gametocytemias to allow for sensitivity and robustness analyses in our results. This 'ensemble modeling' approach has been used previously to model the effects of vaccines on malaria transmission [40] as well as within-host P. falciparum dynamics [41].

Figure 3A illustrates the results of the gametocyte activity model fitting. The untreated model gametocyte prevalence 5 days after first fever is shown in black. The modeled post-treatment gametocyte prevalence assuming a pure schizonticidal combination is shown in green. The four model parameterizations that best correspond to the observed field patterns after CQ treatment are labeled 'CQ mild', 'CQ moderate 1', 'CQ moderate 2', and 'CQ strong', respectively (indicating increasing levels of activity against very early stage gametocytes). The three model parameterizations that best correspond to the ACT field patterns are labeled 'ACT mild', 'ACT moderate', and 'ACT strong', respectively. The four model parameterizations that correspond to the ACT+PQ field studies are labeled 'ACT+PQ moderate <day 0>', 'ACT+PQ moderate <day 2>', 'ACT+PQ strong <day 0>', and 'ACT+PQ strong <day 2>'; the bracketed number indicates the day on which the simulated PQ component was administered, relative to the other two drug components.

Once the gametocyte parameterizations were fitted for each type of drug combination, we then transformed the daily gametocytemias before and after treatment into predicted infectivities to mosquitoes. These transformations utilized gametocyte density-to-infectivity relationships derived from mosquito feeding studies, as described in [23]. We chose two transformations, one derived from studies of mosquito feeding on malaria therapy patients with no prior history of malaria infection ('Jeffery-Eyles' or JE) and the other derived from feeding studies conducted in field trials in Africa ('Carter & Graves' or CG). In short, the JE transformation assumes that 1) gametocytes appearing in the first few days of infection are non-infectious (immature); 2) low density gametocytemias are relatively non-infectious; 3) high density infections are highly infectious. The CG relationship assumes that 1) gametocytes are immediately infectious; 2) low density gametocytemias are relatively infectious; 3) high density infections are not as infectious as for JE [23]. These two functions are substantially different and represent some of the possible types of density-to-infectivity relationships. Calculation using both relationships allows us to highlight where differences in density-toinfectivity assumptions play an important role in interpreting model outputs [23]. It was assumed for both parameterizations that modeled gametocytemias were infectious at densities below the level of detection by microscopy (\sim 5–10 gametocytes per μ L). Our model is thus able to capture the effects of 'submicroscopic' infections [23,35,42]. However, as densities decrease, infectivity decreases asymptotically toward 0 (see Figure 3 of [23]). As a simplification we assumed that gametocyte densities below 2



Figure 3. Modeled post-treatment gametocyte prevalence and treatment effect sizes. Treated and untreated malaria infections were simulated using our within-host malaria infection model. Modeled treatments differed according to the assumed level of gametocyte killing. Model treatment was assumed to start 5 days after the first onset of fever; all model outputs represent the mean of 1,000 runs. (A) The number of individuals predicted to be gametocyte positive by microscopy (threshold 5 gametocytes per µL) was tracked over time. Untreated model outputs are shown in black. Treatment was assumed to be a combination therapy with a short-lived component (active for 3 days) and a longer-lived component with the pharmacokinetic profile of mefloquine. The green line illustrates the effects of treatment assuming no gametocytocidal activity ('Schizonticide'). The assumed gametocytocidal activity of each component was progressively increased and compared to field data to generate the rest of the curves, each labeled with their corresponding antimalarial (chloroquine, CQ; artemisinin-based combination therapy, ACT; primaquine, PQ). The curves labeled 'ACT+PQ' assumed the presence of a second short-lived partner that strongly killed both early and late stage gametocytes; the number indicates the day on which the simulated PQ component was administered. (B) Total effect sizes (fold-reductions in transmission) for each of the modeled drug parameterizations as a function of treatment coverage, including the oocidal effects of drugs, assuming net untreated infectivity of 30.5 days, and using the 'Jeffery-Eyles' density-to-infectivity parameterization for treated individuals (Table 2; [23]). Each drug class is depicted in a different color. The variation in each class is due to the different simulated levels of gametocytocidal activity for that drug type. Each line within a given drug class represents the result of 1,000 simulation runs; the black lines indicate the mean effect sizes for each class of drug. The horizontal line illustrates a six-fold reduction in transmission. The dotted vertical lines indicate the levels of treatment coverage needed to reach a six-fold reduction in total human-to-mosquito transmission for each drug class. The y-axis is in log-scale. doi:10.1371/journal.pcbi.1003434.g003

gametocytes per 3 μ L were non-infectious (given the need for 2 gametocytes per mosquito bite that typically collects ~3 μ L of blood) [23].

Figure S3 illustrates the modeled gametocytemias of **Figure 3A** transformed into probabilities of mosquito infection, along with data from feeding studies in the field [28,30–33], treatment with various ACTs [28,30–32,34–36,38], and treatment with ACTs plus PQ [35,36,38]. Both the JE and CG transformed probabilities are shown; the data transformed using the JE

assumptions are in bold. The coloring of modeled infectivity data corresponds to that of **Figure 3A**. The field feeding study data were disaggregated according to the same criteria as the gametocyte clearance data.

Once the model gametocyte and infectivity parameters were fitted to data, we then calculated the AUIC for each drug parameterization [23]. **Table 1** provides the unadjusted net human-to-mosquito infectivity for each of the drug parameterizations in **Figure 3A**. All data are from the mean of 1,000 model

runs. To determine the transmission reduction achieved with each treatment, we divided the untreated AUIC by the treated AUIC. For example, to calculate the transmission reduction post-treatment, we first took untreated individuals and calculated the mean AUIC for 5 days after first fever until the end of simulation; the mean untreated AUIC values were 31.8 for the JE parameterization and 29.3 for the CG parameterization (**Table 1**). The post-treatment net infectivities of treated individuals, i.e. the treated AUIC values, were 0.70 for JE and 0.94 days for CG (**Table 1**). Mean fold reductions in transmission, post ACT treatment, were then 47.1 and 32.4 for JE and CG transformations, respectively (**Table 1**).

The quantity most relevant for control efforts is the total effect size, i.e., the reduction in transmission that includes the period of transmissibility before treatment. The longer that individuals wait to be treated, the less the maximum effect size achievable, because these individuals could transmit the parasite prior to treatment. After adding pretreatment infectivity (0.03 or 3.06 infectious days for JE and CG transformations respectively), the total mean foldreductions for ACTs were 45.3 and 8.1, respectively. For ACT+PO, total mean effect sizes were predicted to be 92.2 (IE) and 8.9 (CG), respectively. The reason for the large differences between these two transformations is how they incorporate pretreatment infectivity. Thus, pretreatment infectivity plays a major role in the total effect size. For JE, gametocytes were assumed to be non-infectious early in the course of an infection, thus pretreatment infectivity was almost nonexistent and the effect size was determined by post-treatment infectivity. In contrast, the CG model assumed that gametocytes were infectious upon emergence, thus pretreatment infectivity was relatively large compared to posttreatment infectivity.

Purely schizonticidal treatments (with zero gametocytocidal activity) were predicted to reduce post-treatment transmission 6.2 to 5.7 fold (JE and CG, respectively). Including pretreatment infectivity, the mean effect size of a pure schizonticide was 6.2 and 3.9 (JE and CG, respectively). For CQ, which is moderately gametocytocidal against stage I–II gametocytes, total effect sizes were estimated to be 15.6 and 6.0 for the JE and CG parameterizations, respectively. These findings highlight the substantial benefit of drugs with more potent gametocytocidal activity in reducing transmission.

Incorporating effects on mosquito-stage parasite development

The fold reductions in Table 1 illustrate how antimalarials reduce transmission assuming 100% treatment coverage. However, these calculations do not incorporate the oocidal effects of some antimalarials. Treatments that are also oocidal (such as SP, LMF, and MFQ [42]) will have larger effect sizes than predicted in Table 1 because of greater reductions in overall human-tomosquito transmission. To calibrate oocidal drug effects, we compared unadjusted model-predicted infectivity and observed field infectivity post-treatment (see **Text S1** and **Figure S3**). For our ACT model infectivity, at day 7 the feeding studies indicated an infectiousness of approximately 2-3.5%, whereas the model predicted 5–8% under the JE parameterization (see **Figure S3C**). This small difference between model and field studies nearly disappeared by day 14 (possibly because of a dose-response effect of LMF on mosquito stage development). To incorporate oocidal activity, we thus assumed that ACTs (AL or AM) reduced onward infectivity by 50% compared to the mean values in Table 1. Mean post-treatment infectivity values became 0.35 (JE) and 0.47 (CG) net infectious days while the total effect sizes became 83.7 (JE) and 9.2 (CG). These data are shown in Table 2. Again, the

	Pre- treatment	Untreated	Schizont- icide	g					АСТ				ACT+PQ				
Parameterization				mild	mod 1	mod 2	strong	mean	mild	pom	strong	mean	pom <0 <	mod <2>	strong <0>	strong <2>	mean
Net Infectivity (JE)	0.03	31.8	5.12	2.57	2.47	1.88	1.51	2.11	06.0	0.66	0.55	0.70	0.32	0.39	0.15	0.26	0.40
Net Infectivity (CG)	3.06	29.3	5.16	2.81	2.73	2.20	1.78	2.38	1.20	0.89	0.74	0.94	0.48	0.60	0.26	0.44	0.58
Fold-reduction (post- treatment, JE)			6.2	12.4	12.8	16.9	21.1	15.8	35.3	48.1	57.9	47.1	100.3	81.1	208.6	120.2	101.7
Fold-reduction (post- treatment, CG)			5.7	10.4	10.8	13.4	16.5	12.8	24.5	32.8	39.8	32.4	61.3	48.7	112.7	66.5	59.7
Effect size, JE			6.2	12.3	12.7	16.7	20.7	15.6	34.3	46.3	55.4	45.3	92.8	76.2	178.4	109.6	92.2
Effect size, CG			3.9	5.5	5.6	6.2	6.7	6.0	7.6	8.2	8.5	8.1	9.2	8.8	9.8	9.3	8.9
Abbreviations: JE, Jeff transmission potential. T treatment. The latter cha ACT the simulated game was administered	ery-Eyles; CG, Ca his value indicat nge is known as tocytocidal com	rter & Graves; C ces the mean nu the 'effect size' (ponents were m	.Q, chloroquine; Imber of days th of an interventioi iild, moderate, ai	ACT, arter lat an ind n. The mo nd strong.	misinin-ba ividual is i deled para . For ACT+	sed combinati nfectious for a ameterizations PQ the param	on therapy; a mosquito v for CQ are n ieterizations	PQ, primac ector. Folc nild, moder are modera	luine. Net -reductio ate 1, mo ate and st	infectivit ns indicat derate 2, rong; the	:y indicates :e the fold and strong numbers i	the area u changes in (increasing n brackets i	nder the ir net infecti the assum ndicate th	nfectivity of vity either ed gametr e treatmer	curve, a me post-treat ocytemic po nt day on w	asure of the ment or pre otency of tre hich the PC	e malaria + and post- eatment). For ! component

Table 1. Modeled net infectivities and effect sizes of antimalarial treatment

doi:10.1371/journal.pcbi.1003434.t00

assumed importance of pretreatment infectivity played a crucial role in determining the effect sizes of treatment when the oocidal effects of treatment were included.

When examining the effects of ACTs plus PQ, our adjustment for the oocidal effects of PQ was different than that for LMF or MFQ, because PQ is active against mosquito stages for only a few days after treatment, but reduces infectivity almost completely during its period of activity [43,44]. If we assumed that infectivity in the first three days post ACT+PQ treatment was zero, then the net infectivity post-treatment was 0.19 (JE) and 0.30 (CG) and the total mean effect sizes were 162.1 (JE) and 9.6 (CG), respectively.

Figure 3B illustrates the total effect sizes for each of the modeled drug parameterizations as a function of treatment coverage, including the oocidal effects of drugs from **Table 2**, assuming that the mean period of infectivity in untreated individuals is 30.5 days (mean untreated infectivity from **Table 1**) and assuming the JE density-to-infectivity parameterization for treated individuals. Each drug class is depicted in a different color with individual lines showing simulation outputs assuming varying levels of gametocytocidal activity (CQ: mild, moderate 1, moderate 2, strong); (ACT: mild, moderate, strong); (ACT+PQ: moderate 0 days delay, moderate 2 days delay, strong 0 days delay, strong 2 days delay). The black lines clustered within each drug class indicate the mean effect size across all simulations. The horizontal line illustrates a sixfold reduction in transmission, a threshold discussed below. The dotted vertical lines indicate the levels of treatment coverage needed to reach the six-fold reduction in total human-to-mosquito transmission for each drug class. The y-axis is in log-scale. This figure graphically illustrates the importance of treatment coverage in determining the effect size of a control program.

At 100% coverage (i.e. all infected individuals), the effect size of ACTs is 87.3 (JE parameterization, **Table 2**), assuming untreated infectivity is 30.5 days. However, this value drops quickly, yielding 16.3 at 95% coverage, 9.0 at 90% coverage, 4.8 at 80% coverage, and 3.2 at 70% coverage. For ACT+PQ, the effect sizes are 162.1, 17.7, 9.4, 4.9, and 3.3 at 100%, 95%, 90%, 80%, and 70% coverage, respectively. Treatment with ACTs exceeds a six-fold reduction threshold at ~84.3% coverage of the total population, whereas the ACT+PQ regimen exceeds a six-fold reduction at ~83.9% coverage of the total population.

Sensitivity analyses

The above effect sizes (and those in **Figure 3**) were calculated assuming the untreated infectivity is 30.5 days (the mean of the JE

and CG parameterizations) and using the JE parameterization for treated individuals. If we use the average of the JE and CG parameterizations for treated individuals, and still assume that the untreated net infectivity is 30.5 days, the pretreatment net infectivity is 1.55 (average from Table 1), the post-treatment net infectivity for ACTs is 0.41 (average from **Table 2**), and the computed effect sizes of ACTs become 15.6 at 100% coverage, 9.0 at 95% treatment coverage, 6.3 at 90% treatment coverage, 4.0 at 80% coverage, and 2.9 at 70% coverage. Assuming a posttreatment net infectivity of 0.245 for ACT+PQ (average from Table 2), these values become 17.0, 9.4, 6.5, 4.0, and 2.9 at the coverage levels listed above. Treatment with ACTs reaches a fivefold reduction in transmission at 85.5% coverage of the total population; with ACTs+PQ, the coverage level required is 85.0%. Treatment with ACTs exceeds a six-fold reduction threshold at 89.1% coverage, whereas the ACT+PQ regimen exceeds a six-fold reduction at 88.5% coverage. These outputs suggest a barely detectable impact of adding PQ to ACTs in the context of reducing transmission levels with these model assumptions.

Mapping of fold-reductions in transmission necessary for malaria elimination

To contextualize the fold-reductions in transmission theoretically achievable with various treatments, we developed a set of maps of the fold-reductions in malaria transmission necessary to achieve elimination in low-transmission settings. These maps were derived from the worldwide maps of the basic reproductive number of malaria, R_0 [45], assuming the malaria control coverage of 2010 as the baseline. We can also consider these maps as calculating the $R_{\rm C}$, i.e. the reproductive number under control efforts, as of 2010, though here we use the terms R_0 and $R_{\rm C}$ interchangeably. R_0 is a threshold criterion for transmission: if $R_0>1$ over a given region, the disease will spread within this region (unless there is significant migration), if $R_0<1$, the disease will disappear within this region (unless there is significant importation).

In brief, the R_0 values described in [45] were developed by regressing various malariometric data (such as elevation and rainfall) on tens of thousands of parasite rate surveys and modeling the spatio-temporal autocorrelation structure of the residual variation. The regressions used were Bayesian and geostatistical, producing a full Bayesian posterior distribution for the agestandardized parasite rate at each pixel (with a per-pixel size of 5×5 km, i.e. 5 km²).

Table 2. Modeled net infectivities and effect sizes of antimalarial treatment (adjusted for oocidal effects).

	АСТ				ACT+PQ				
Parameterization	mild	mod	strong	mean	mod <0>	mod <2>	strong <0>	strong <2>	mean
Net Infectivity (JE)	0.45	0.33	0.27	0.35	0.24	0.26	0.10	0.16	0.19
Net Infectivity (CG)	0.60	0.45	0.37	0.47	0.34	0.41	0.16	0.29	0.30
Fold-reduction (post-treatment, JE)	70.5	96.2	115.9	94.2	129.7	120.4	318.5	194.6	190.8
Fold-reduction (post-treatment, CG)	49.0	65.6	79.6	64.8	86.3	71.3	180.1	99.7	109.4
Effect size, JE	66.8	89.3	106.0	87.3	117.4	109.7	253.1	168.1	162.1
Effect size, CG	8.9	9.2	9.4	9.2	9.5	9.3	10.0	9.7	9.6

Abbreviations: JE, Jeffery-Eyles; CG, Carter & Graves; ACT, artemisinin-based combination therapy; PQ, primaquine. Net infectivity values indicate the mean number of days that an individual is infectious for a mosquito vector, as explained in **Table 1**. Effect sizes illustrate the fold reductions in transmission for an ACT or ACT+PQ treatment, modeled with distinct levels of gametocytocidal activity). The numbers in brackets for the ACT+PQ treatments indicate the treatment day on which the PQ component was administered.

doi:10.1371/journal.pcbi.1003434.t002

These worldwide maps of predicted parasite rates varied in intensity from pixel to pixel, given different magnitudes of various malaria covariates. By utilizing empirical and theoretical relationships, the maps were combined with aspects of malaria that remain constant over time and space to calculate R_0 . One such malaria invariant is the net infectivity of infected humans to mosquitoes, assuming no acquired immunity developed over the course of repeated infections [23]. To calculate this invariant, we used our within-host model of malaria transmission to simulate the progression of infectivity in thousands of simulated individuals, and then calculated the mean area under the human-to-mosquito infectivity curves [23].

To achieve elimination in a given area, the fold-reduction in transmission under control must be greater than or equal to the R_0 [46]. Thus, we took the worldwide maps of R_0 in [45] and calculated the fold-reductions necessary over each pixel to reduce the estimated $R_{\rm C}$ to below 1. Our maps of the transmission reductions include estimates of uncertainty inherited from the $R_{\rm C}$ posterior densities at each pixel. **Table 3** summarizes these maps by providing the number of people living in areas requiring <2 fold reductions in transmission to interrupt transmission, as well as 2–5, 5–10, 10–20, 20–50, 50–100, and >100 fold reductions to interrupt transmission. **Table 3** calculates these necessary fold-reductions with 75% confidence, i.e., our posterior estimates of $R_{\rm C}$ fall within the given regions with 75% confidence.

Because our within-host model simulates the progression of infections for individuals with no prior history of malaria infection, our modeling conclusions are most relevant for areas of low transmission where individuals have accumulated little acquired immunity from prior infections. We thus restricted our analyses to areas with $R_0 < 10$, which excludes the more endemic areas of Africa and aligns with the transmission levels prevalent in Southeast Asia. In terms of biting intensity, an R_0 of 10 translates to a yearly entomological inoculation rate of approximately 3 infectious bites per person per year, which would result in approximately 1.5 infections every year [47]. This upper limit of analysis can be compared with the intensity reported in an area of 'low and seasonal' transmission in Thailand, where individuals had one infection every other year [34].

Figure 4 provides a worldwide map of the probabilities that areas can interrupt malaria transmission ($R_C < 1$) assuming a five-fold reduction in transmission. Areas with $R_0 > 10$ are masked, as these regions have such high transmission that our modeling predictions are less relevant. Fold reductions were organized into 6 bins for clarity. **Figure 5** shows these probabilities of interruption for Southeast Asia, where transmission is generally much lower than in Africa. **Figures S4** and **S5** present maps of the probabilities of interrupting transmission assuming two- or tenfold reductions in transmission, respectively. These maps are discussed below in the context of the reductions achievable with antimalarial drugs.

Discussion

This study calculates the effects of antimalarial therapies on *P. falciparum* transmission, using a within-host model of malaria infection [23] and a PK/PD model parameterized from field studies. The effects of drugs are modeled on both asexual and sexual stages of parasite development under different assumptions

Table 3. Required effect size required to reduce R_{C} below 1, worldwide and solely in Southeast Asia, stratified by populations at risk.

				Total populat	ion
	Required effect size	Population	Population at risk (%)	(%)	
Worldwide	no risk	4,362,130,000		62.89%	
	<2	1,672,204,000	64.97%	24.11%	
	2–5	223,901,000	8.70%	3.23%	
	5–10	94,173,000	3.66%	1.36%	
	10–20	83,986,800	3.26%	1.21%	
	20–50	123,546,000	4.80%	1.78%	
	50–100	84,695,100	3.29%	1.22%	
	>100	291,346,000	11.32%	4.20%	
Total population					6,935,981,900
Total pop at risk					2,573,851,900
Southeast Asia	no risk	56,595,000		24.98%	
	<2	135,508,000	79.71%	59.80%	
	2–5	19,389,600	11.41%	8.56%	
	5–10	7,800,520	4.59%	3.44%	
	10–20	3,838,410	2.26%	1.69%	
	20–50	2,955,060	1.74%	1.30%	
	50-100	477,035	0.28%	0.21%	
	>100	39,566	0.02%	0.02%	
Total population					226,603,191
Total pop at risk					170,008,191

Abbreviations: Southeast Asia is defined here as encompassing the countries Thailand, Myanmar, Laos, Vietnam, and Cambodia. Required effect size denotes the foldreduction in malaria transmission needed to interrupt transmission over the given population. Confidence of interruption is at the 75% level. doi:10.1371/journal.pcbi.1003434.t003



Figure 4. Worldwide map of the predicted probabilities that a five-fold effect size will interrupt malaria transmission. This map shows the predicted probabilities that a five-fold reduction in transmission ('five-fold effect size') would interrupt malaria transmission over a given pixel. Map pixel size is 5 km². In order to interrupt malaria transmission in a given area, the basic reproductive number for malaria under control (R_c) needs to be reduced below 1. Probabilities for each pixel are calculated according to Bayesian posterior estimates of uncertainty [45]. Probabilities have been binned into six categories for clarity. Areas with high transmission (defined as at last a 50% probability of R_0 >10) are masked because our model results are applicable to regions of relatively lower transmission. Most of Sub-Saharan Africa is masked because of the very intense transmission. However, most of Sahelian Africa, as well as East Africa, parts of Southern Africa, most of India, as well as most of Southeast Asia and essentially all of South America have high probabilities of interruption at this control level. Note that local conditions (within a given pixel) may be more or less favorable to transmission than the per-pixel averages shown here, and so these maps are most applicable for regional or country-level planning, rather than local-level control efforts. Microenvironments or 'hotspots' might require additional interventions and/or greater treatment coverage than the per-pixel average [67]. Data collection sites used to construct the maps are reported in [45]. A partial database of the actual site locations and the measured levels of malaria endemicity can be found on the Malaria Atlas Project (MAP) website: www.map.ox.ac.uk. doi:10.1371/journal.pcbi.1003434.g004

of gametocyte-to-infectivity relationship. We also generate global maps of the fold reductions in malaria transmission, i.e. the effect sizes, necessary to achieve elimination in regions of low endemicity (defined as having local R_0 values less than 10).

From our model outputs, we can generate three major conclusions. First, the infectivity of individuals before treatment plays a crucial role in determining effect size. If treatment is delayed more than only a few days after the onset of fever, and gametocytes are infectious during this period, then the effect sizes achievable even with first-line ACT therapies plus the gametocytocidal agent PQ are limited. Second, if we account for the effects that the partner drugs LMF and MFQ exert upon mosquito stages of the



Figure 5. Map of the predicted probabilities that five-fold reductions will interrupt transmission in Southeast Asia. The predicted probabilities that a control effort with a five-fold reduction would interrupt transmission are shown for Southeast Asia, using the same masking of high transmission areas (R_0 >10) and mapping assumptions as for **Figure 4**. Areas that appear to be uniform may have small-scale heterogeneities in transmission that are beyond the scale of this map. Map pixel size is 5 km². doi:10.1371/journal.pcbi.1003434.g005

parasite life-cycle, then there is little difference in the benefits of ACTs versus ACTs+PQ in terms of transmission reductions. Both regimens are extremely effective at stopping onward transmission, with many fold greater benefits versus purely schizonticidal treatments that act only upon asexual blood stage parasites. Third, the proportion of individuals receiving treatment has a major impact on reductions in transmission (**Figure 3B**). In **Tables 1** and **2** our effect size calculations assumed 100% coverage. Because untreated individuals are so much more infectious than treated individuals, leaving even a few individuals untreated drastically reduces the effectiveness of a control program.

We can put these fold-reductions in context using our maps of transmission reductions necessary for elimination. **Figure 4** illustrates a worldwide map of the probabilities that a five-fold reduction in transmission would interrupt the spread of malaria. In this map the pixel size is 5 km². Because our model is most applicable in regions of relatively low transmission, we masked out the regions where R_0 , the basic reproductive number, is predicted to be greater than 10 (with a probability exceeding 50%). Higher transmission regions are more difficult to model, given the complex interactions of immunity, superinfection, and control. As can be seen from the map, many areas of Africa have such intense transmission that R_0 exceeds 10, and we cannot say how transmission might be affected by the use of drugs in such areas.

However, examining the map, one can visualize many regions of Africa, including the Sahel, most of East Africa, and parts of Southern Africa, where elimination would appear possible with a five-fold reduction in transmission. Further, much of India and Southeast Asia have low enough transmission that elimination would be possible at this level of control. Prospects for elimination in Myanmar and southern Thailand, however, do not appear to be favorable. Figure 5 provides a zoomed-in view of the probabilities of malaria elimination with an effect size of five, focusing on Southeast Asia. Table 3 provides a quantification of the populations at risk both worldwide and in Southeast Asia, as well as the populations where elimination is possible at different levels of control. Worldwide, regions where malaria can be interrupted with five-fold reductions contain 74% of the population at risk; in Southeast Asia regions that can interrupt transmission with fivefold reductions harbor 91% of the population at risk.

Given these maps and quantifications of populations at risk, we can apply our modeling results to determine the percentage of the population that needs to be treated promptly with antimalarials to interrupt transmission in various areas. Figure 3B illustrates the relationship between treatment coverage with different antimalarials and the resulting fold reductions in transmission, assuming the Jeffery-Eyles gametocyte density-to-infectivity relationship (used to calculate the needed treatment coverage levels below). To achieve a five-fold reduction in transmission, approximately 81% of the total infected population would need to be treated with ACTs or ACTs+PQ. As a comparison, achieving this fold reduction with CQ (with weak activity against early-stage gametocytes) or a schizonticide (with no gametocytocidal activity) would require treatment coverage of $\sim 86\%$ and $\sim 96\%$ respectively. To achieve a six-fold reduction in transmission, approximately 84% of the total infected population would need to be treated with ACTs or ACTs+PQ (illustrated in Figure 3B). Our modeling thus suggests that the addition of PQ to an ACT would provide almost negligible benefits at these levels of coverage, reducing the fraction of the population needing to be treated by less than 1% versus treatment with ACTs alone that already provide quite potent gametocytocidal activity (Figure 3B).

Combining the maps and the within-host modeling results based on our JE parameterization, we thus estimate that promptly treating $\sim 81\%$ of the total infected population with ACTs and/or ACTs+PQ would interrupt transmission in areas covering 91% of the population in Southeast Asia. These coverage rates are for the infected population as a whole, regardless of whether individuals are symptomatic or not. In a study conducted in a region of western Thailand with low and seasonal transmission, most infections (87%) were found to be symptomatic [48]. In experimental challenge studies among human volunteers, all subjects displayed some degree of symptoms [49].

If we take the former proportion (87%) as the percentage of the infected population that is symptomatic, then 93% of the symptomatic population would need to be treated with ACTs and/or ACTs+PQ to achieve interruption in the areas of Figure 5 based on the predicted five-fold reduction; the percentage rises to 97% to achieve six-fold reductions. Thus, it is possible that treating only symptomatic individuals may be sufficient to eliminate transmission throughout most of Southeast Asia. Almost all of these individuals, however, would need to be reached with treatment (either through a public campaign or private sector provisioning or a combination of both) in order to interrupt transmission, using drugs alone. Our result that prospects for malaria elimination are favorable for most of Southeast Asia is supported by two other studies that also find that elimination efforts are feasible using antimalarials in this region [50,51]. Figures S4 and S5 illustrate the probabilities assuming control interventions with two-fold and ten-fold reductions, respectively. At the two-fold level, much of central Thailand can interrupt transmission, but there are significant portions of Myanmar, Cambodia, Southern Laos, and Southern Thailand where elimination is not likely. At the ten-fold level, there are small pockets in Southern Thailand as well as large areas in Myanmar where interruption is still not likely.

These estimates may be somewhat optimistic because we are using only the JE density-to-infectivity relationship when calculating the infectivity of treated individuals. As we are focusing on low transmission areas, this assumption seems reasonable (the JE relationship was derived from individuals with no prior infections). However, if we take the mean of the JE and CG relationships for treated individuals, then we predict that it would require treating 85.5% of the total population, or 98.3% of symptomatic patients with ACTs to achieve a five-fold reduction (85.0% of the total or 97.7% of symptomatic patients with ACTs+PQ). Further, we find that it is not possible to achieve a six-fold reduction only treating symptomatic individuals with either the ACTs or the ACTs+PQ modeled here, assuming the mean of the JE and CG relationships for treated individuals. Thus, the assumed density-to-infectivity relationship has a large effect on the calculated effectiveness of control programs.

We note that our maps predict the levels of control necessary to interrupt transmission at the per-pixel level (5 km²), incorporating uncertainty analysis. These average reductions needed to interrupt malaria transmission are not at the per-village or per-household level. Hotspots of transmission will need to be identified and treated in order to achieve elimination in a given region. The uncertainty for each pixel takes into account this heterogeneity to some degree, but nevertheless caution is advised before using these maps for local-scale planning. We would suggest our maps be used to guide elimination planning at a regional or national level; for elimination planning at a district or city level more intensive surveillance will likely be needed.

If we consider the timelines to elimination, the narrower the margin by which the effect size exceeds the threshold for elimination, the longer elimination will take, as population-wide transmission will decay more slowly [46]. Conversely, the higher

the proportion of individuals above the needed threshold, the faster elimination will be achieved [46]. We do not compute the quantitative benefits of mass drug administration here, and instead focus on individual-level treatment. However, we would qualitatively expect that mass drug administration may provide a benefit to elimination efforts by speeding an area toward faster elimination, assuming that the critical level of coverage can be reached (i.e. the $R_{\rm C}$ of the region drops to below 1). Once an area has eliminated malaria, the costs of maintaining elimination may be less than those needed to achieve elimination in the first place, though more research is needed on strategies to maintain elimination in previously endemic areas [7].

In areas where antimalarials are predicted to be insufficient to achieve elimination, other interventions may be included in control efforts to increase the effect size of the combined control effort. The combined effect size is simply the product of both component interventions. For example, if the coverage level with antimalarials reduces transmission by three-fold and distribution of bed nets reduces transmission another three-fold, the combined effects are a nine-fold reduction in transmission (as long as there are no antagonistic interactions between the two efforts). Thus, high fold-reductions can be achieved by bundling interventions. While we do not compute the effect sizes of other interventions here, the results in this paper can be combined with other modeling efforts for the purposes of an integrated elimination effort.

Given these conclusions, serious efforts to eliminate malaria will require extensive planning and sustained support [6]. We note the encouraging prediction that high coverage (at least 81-85% of total infections, corresponding to an estimated 93-98% of symptomatic infections) with ACTs that act against P. falciparum asexual, sexual, and mosquito stages might suffice to interrupt transmission throughout most of Southeast Asia, especially if complimented by insecticide-treated bed net distribution to reduce population infectivity. We also note that the addition of a single dose of a purely gametocytocidal drug such as PQ to ACTs can reduce onward transmission slightly. However, the focus of control efforts should be on maintaining a high level of treatment coverage. Based on our modeling, PQ and similarly gametocytocidal therapies added to ACTs do not appear to be a magic bullet ensuring elimination and add only nominally to the transmission reductions achievable with ACTs that act against the various parasite stages at feasible levels of coverage.

Efforts are ongoing to utilize our model to predict the effects of possible emerging artemisinin resistance, which threatens existing ACT control strategies [10,52,53]. Additional modeling is also required to delineate better what measures beyond expanded ACT and insecticide-treated bed net coverage would help reduce the $R_{\rm C}$ in Africa to levels that might make elimination an achievable goal.

Materials and Methods

Model description

Our recently reported within-host mathematical model [23] was utilized to simulate asexual and sexual blood stage parasite densities over time in untreated and treated individuals. This model reproduces the range of observed parasite densities among individuals undergoing malaria therapy, wherein adult males with tertiary syphilis (and no acquired immunity to malaria) were infected with various strains of *P. falciparum* to induce a fever in order to clear the syphilis [29,54–56]. Our model uses a combination of parasite antigenic variation and host immune responses to reproduce the observed range of responses in these patients. The model calculates the density of asexual parasites every two days and uses log-linear interpolation to generate daily counts. The model also calculates the daily human-to-mosquito infectivity using gametocyte density-to-infectivity relationships derived from mosquito feeding studies on human volunteers [23]. The source code for our model is provided in **Dataset S1** (see **SI**). We note that our modeling uses discrete-time difference equations rather than a continuous time model, to calculate both asexual densities and gametocyte densities over time. We chose the former, as the calculation of gametocyte densities from asexual densities is difficult with a continuous-time model because gametocyte densities are a function of weighted cumulative asexual densities and are highly stochastic. For a thorough description, we refer to [57].

We also note that an insightful report by Kay and Hastings [58], building on earlier work from this group [59], simulated the concentrations of both artemisinin derivatives and ACT partner drugs using compartmental PK/PD modeling. These authors also simulated the killing effects of these drugs against asexual parasites assuming Michaelis-Menten dose-response functions (which are similar to the Hill functions used here). These authors focused on the effects of emerging artemisinin resistance and provided substantial data on artemisinin PK/PD properties. Our two studies differ in that we simulated parasite densities daily and so we did not explicitly model artemisinin PK values because of the short half-lives of the artemisinins. Further, our stochastic difference equation model of within-host parasite growth and immune response was calibrated to match the range of variation in the malaria therapy data [23]. We also modeled the effects of drugs against asexual, sexual, and mosquito stages of development. Thus, The Kay and Hastings study [58] is a representation of the effects of drugs against asexual stages incorporating resistance, whereas our study focused on the effects of drugs on total human malarial transmission. Importantly, their simulations predict that the spread of ART resistance would result in a potentially rapid decline in ACT effectiveness. In future studies, we plan to investigate the effect of ART resistance on effect sizes and how this would impact the required treatment coverage to drive the $R_{\rm C}$ to below 1 in low-transmission settings.

Antimalarial drug pharmacokinetics

PK modeling was used to simulate the concentrations of antimalarial drugs after uptake. The concentrations of CQ and its active metabolite, monodesethyl-chloroquine (mdCQ), were simulated using a non-compartmental model parameterized with data from Papua New Guinean children [60]. For LMF, a twocompartmental model was developed from plasma concentration data from the treatment of uncomplicated individuals in western Thailand (Mae La) [61]. For MFQ, a non-compartmental model was developed that incorporated data from two studies, one in Thailand [62] and the other in Peru [63] (the Peruvian study used whole blood concentrations, rather than plasma). The plasma concentrations of the artemisinins were not modeled, although such data exist [62]. This is because the half-lives of the artemisinins are so short ($\sim 1-3$ hr) that effective concentrations are gone within one day after uptake [62]. A full description of the PK modeling is provided in the SI.

Drug pharmacodynamics against asexual parasites

The dose-response effect of antimalarials against as exual parasites was assumed to follow the commonly used 'Hill function' [64], a four-parameter dose-response function: $K(x) = \left(\left(a - b \right) / \left(1 + \left(x/c \right)^d \right) \right) + b$, where *a* is set to 0, *b* is set to 1, *c* is what we term the EC_{50} , d is the Hill slope, and x is the plasma concentration of the drug. The Hill function dose-response curve type was chosen because this function type was utilized by both of the references that provided in vivo EC_{50} values for MFQ [65,66], and because many of the in vitro studies using CQ and LMF used Hill dose-response relationships to model the effects of drugs against asexual blood stage parasites. Because each drug has a characteristic maximum inhibitory effect, this dose-response function was scaled by the maximum parasite reduction ratio (PRR) for each drug. To determine the effect on asexual parasites of a drug concentration on a given day t of modeling, the asexual parasite densities from day t-1 were used as inputs into the withinhost model. The predicted densities on day t were then calculated, incorporating the effects of host immunity and parasite growth. The mean of $\sqrt{PRR} \cdot K(x)$ over day *t* was then subtracted from the within-host simulations after appropriate log transformation to calculate the end of day asexual parasite densities, incorporating the effects of host immune responses, parasite growth, and drug concentrations. These densities were then used to calculate the asexual parasite densities on day t+1, and so on until the end of the simulation time. A full description of the PD modeling against asexual blood stages is provided in the SI.

Drug pharmacodynamics against gametocytes

The dose-response effect of antimalarials against gametocytes was assumed to follow a binary model, where antimalarials act against gametocytes only if drug concentrations are above a certain drug-specific threshold. This binary model was adopted because of the current paucity of dose-response data against gametocytes, as compared to asexual blood stage parasites where the many data sets permit the use of Hill slopes to define doseresponse relationships. The threshold for gametocyte activity was chosen to be the in vitro IC50 against asexual blood stage parasites scaled by a factor of five [21]. For CQ, LMF, and MFQ, this value is 40, 174, and 322 ng/ml, respectively (SI). To determine the stage-specific gametocytocidal effects of drugs, the within-host malaria model was first run assuming treatment with a purely schizonticidal combination therapy. The post-treatment gametocyte clearance curves from these simulations were then compared to clearance curves from field studies using SP [28,30-38] to validate the model outputs. In separate simulations, we assumed that individuals were treated with a combination that was weakly gametocytocidal, and these results were compared to field data from CQ trials [28,30-33] to choose a drug parameterization. We also performed simulations assuming treatment with a stronger gametocytocidal combination, and modeled our parameters by comparing these results with ACT field trial data [28,30-32,34-36,38]. Finally, we simulated treatment with a stronger gametocytocidal combination paired with a third highly gametocytocidal drug, and compared these results to ACT+PQ field data [35,36,38] to parameterize this combination. Outputs from modeled drug parameterizations that were consistent with observed trends were considered representative of that type of treatment. A full description of the process of model parameterization of antimalarial effects against gametocytes is provided in the SI.

Drug pharmacodynamics against mosquito-stage parasites

Once the ensembles of gametocyte densities after treatment had been generated for various drug combinations, we used two different gametocyte density-to-infectivity relationships ('Jeffery-Eyles' and 'Carter & Graves') to translate the daily gametocyte densities into predicted human-to-mosquito infectivities [23]. These modeled daily infectivities were then compared to field studies in which mosquitoes were fed on human volunteers after treatment. For the effects of a single dose of PQ on drug transmission, we assumed that the first three days post-treatment were non-infectious (including the day of treatment). For the effects of partner drugs with longer half-lives that are active against sexual-stage parasites for longer periods (LMF, MFQ), we scaled the area under the infectivity curve (AUIC).

Analysis of model uncertainty

To quantify the uncertainty associated with our predictions, we utilized an ensemble modeling approach [40,41]. In ensemble modeling, various scenarios are simulated to illustrate the effects of changing assumptions of model outputs. Ensemble modeling is especially appropriate when insufficient or conflicting data exist to determine the relative likelihoods of the possible scenarios. For our ensembles, we used different sets of assumptions about the stagespecific effects of drugs against gametocytes and the type of gametocyte density-to-infectivity relationship to map out the uncertainties associated with our best-estimate predictions of the effects of drugs on transmission.

Supporting Information

Dataset S1 Text of source code for model (final_model.rtf).

(RTF)

Figure S1 Post-treatment gametocyte prevalences from field studies. The graphs show gametocyte prevalences of field populations after antimalarial treatment. Gametocyte positivity was assessed using microscopy (threshold \sim 5–10 gametocytes per µL blood). The notation ('- Day 0') indicates that only individuals who were gametocyte negative at admission were included in the study. Field notes include the location of study and subset of population treated. All study curves represent mean population values linearly interpolated from measured prevalences. $(\!A\!)$ The percentage of individuals positive for gametocytes after treatment with sulfadoxine-pyrimethamine (SP) [28,30,33]. The pattern after SP treatment can be described as an inverted-V: few mature gametocytes were present at treatment because treatment was relatively prompt and some studies excluded gametocyte carriers at admission. A peak in prevalence was caused by sequestered gametocytes emerging into the blood stream. The peak gradually declined as the immune system cleared gametocytes from the blood. (B) Gametocyte prevalences after treatment with chloroquine (CQ) or amodiaquine (AQ) (sometimes in combination with SP) [28,30–33]. (C) Gametocyte prevalences after treatment with various artemisinin-based combination therapies (ACTs): (artemether-lumefantrine, AL), (artesunate, A1, A3, AS), (artesunatemefloquine, AM), (dihydroartemisinin-piperaquine, DHP) [28,30-32,34–36,38]. (D) Gametocyte prevalences after treatment with ACTs plus primaquine (PQ) [35,36,38]. (PDF)

Figure S2 Comparison of modeled post-treatment gametocyte prevalences to field study data. The posttreatment gametocyte prevalences from Figure S1 were averaged to create a set of target data to parameterize the modeled effects of antimalarials on transmission. (A) The mean of the field data after treatment with sulfadoxine-pyrimethamine (SP) [28,30,33] is illustrated by the red line and the range of observed responses are depicted in light blue. The modeled gametocyte carriage curves are shown in black and green. All model treatment was assumed to start 5 days after the first onset of fever, consistent with average behavior from field studies [34,38]. All model outputs represent the mean of 1,000 runs. The solid black line illustrates modeled gametocyte carriage among untreated individuals; the dashed black line illustrates modeled clearance in untreated individuals among gametocyte negatives at admission ('- Day 0'). Modeled gametocyte clearance in untreated individuals is mediated only by immune processes as described in [23]. The solid green line depicts modeled gametocyte prevalence after treatment with a schizonticidal combination therapy (i.e. a shortlived component that rapidly kills asexual parasites and a longerlived one that is less potent; neither are assumed to affect gametocytes); the dashed green line depicts modeled gametocyte carriage after schizonticidal treatment, including only gametocyte negative individuals at treatment ('- Day 0'). (B) Gametocyte prevalences after treatment with chloroquine (CQ) or amodiaquine (AO) (sometimes in combination with SP) [28,30-33]; mean values are illustrated in red, range in blue. Modeled gametocyte prevalences are also provided. Model prevalences assume treatment with a combination of drugs (short-lived and long-lived) that kill asexual parasites but only affect early stage gametocytes. Mild, moderate (2 being stronger than 1), and strong model outputs vary in the assumed intensity of early stage gametocyte killing. (\mathbf{C}) Gametocyte prevalences after treatment with various artemisinin-based combination therapies (ACTs) [28,30-32,34-36,38]; mean values are in red and the range is illustrated in blue. Model data illustrate treatment with a combination of drugs, a shorter-lived one that kills both early and late stage gametocytes and a longer-lived partner that only kills early stages. Model outputs vary in their assumed gametocyte killing strengths. The curve labeled 'no late stage effects' illustrates the effects of a combination that has no effect on gametocytes aged >13 days old. (**D**) The same data as in (**C**) are illustrated with the exception that two ACT field studies were removed from the field data curves because of potentially confounding drug resistance effects (CQ+AS, [31] and SP+AS, [35]). (E) Gametocyte prevalences after treatment with ACTs plus primaquine (PQ) [35,36,38]; the red line indicates the mean of field studies; the range of studies is shown in blue. Model outputs assume treatment with three drugs: a short-lived drug that kills early and late stage gametocytes, a long-lived partner that kills late stage gametocytes only (these first two drugs were parameterized from the ACT field data, using the 'mild' parameterization), and a second short-lived partner drug that strongly kills both early and late stage gametocytes. The designation of $\langle day 0 \rangle$ or $\langle day 2 \rangle$ refers to the numbers of days that single dose PQ was delayed after initiating ACT treatment (i.e. simultaneously or 2 days after). (PDF)

Figure S3 Infectivity to mosquitoes after treatment. These graphs illustrate the probability that a human will infect a mosquito following antimalarial treatment. Infectivity is defined as the probability that a mosquito bite will produce oocysts. Field study data are indicated with markers; model outputs are indicated by curves. All model outputs represent the mean of 1,000 runs; treatment was assumed to begin 5 days after first fever. Field markers represent the mean from a set of mosquito feedings. Model output curve coloring is taken from **Figure S1**. Two different gametocyte density-to-infectivity relationships were used to model infectivity: Jeffery-Eyles (in bold; JE) and Carter & Graves (CG) [23]. Some field and model data included only gametocyte negative individuals at admission, as indicated by ('Day 0 -'); others included all individuals ('Day 0 + / -'). (A) Field data post-treatment with SP (sulfadoxine-pyrimethamine) or SP plus amodiaquine (AQ) [28]. The modeled outputs are from simulations approximating the effects of chloroquine (CQ) treatment. Field-measured infectivity after SP treatment resembles that of modeled CQ treatment, even though gametocyte densities after SP treatment were much higher than after CQ. The discrepancy is explained in part by evidence that SP acts against the mosquito stages of development, thus reducing the human-to-mosquito infectivity for given levels of gametocytemia [68–70]. (B) Field-measured infectivity after treatment with CQ, SP, or CQ+SP [30-33]. Model outputs were normalized to remove simulated individuals positive at treatment. The JE parameterization is more consistent with field data, although it is unclear how infectious individuals were 0-3 days post-treatment. (C) Field-measured infectivity [28,37] and modeled outputs after treatment with artemisinin-based combination therapies (ACTs), including only individuals that were gametocyte negative at admission. (D) Field-measured infectivity [30-32] and modeled outputs after treatment with artemisinin-based combination therapies (ACTs), including all treated patients. The JE parameterization is more consistent with field data, at least after the initial period of infection. (E) Field-measured infectivity after treatment with ACTs [28,37]; modeled infectivity is for ACTs plus primaquine (ACT+PQ). The lack of field infectivity data for the ACT+PQ combination treatment precludes direct comparison, but the ACT-treated field data is provided for reference. (\mathbf{F}) Field-measured infectivity after treatment with ACTs, excluding individuals positive for gametocytes [30-32]; modeled infectivity is for ACT+PQ, also after exclusion. The JE parameterization produced infections that are infectious only at an extremely low level.

(PDF)

Figure S4 Maps of the predicted probabilities that a two-fold effect size will interrupt malaria transmission. The upper map shows the predicted probabilities that a two-fold reduction in transmission ('two-fold effect size') would interrupt malaria transmission over a given pixel. Map pixel size is 5 km². In order to interrupt malaria transmission in a given area, the basic reproductive number for malaria under control $(R_{\rm C})$ needs to be reduced below 1. Probabilities for each pixel are calculated according to Bayesian posterior estimates of uncertainty (45). Probabilities have been binned into six categories for clarity. Areas with high transmission $(R_0 > 10)$ are masked because our model results are applicable to regions of relatively lower transmission. Note that local conditions (within a given pixel) may be more or less favorable to transmission than the per-pixel averages shown here, and so these maps are most applicable for regional or country-level planning, rather than local-level control efforts. Microenvironments or 'hotspots' might require additional interventions and/or greater treatment coverage than the pixel average [67]. The lower map inset illustrates the predicted probabilities that a control effort with a two-fold reduction would interrupt transmission in Southeast Asia, using the same masking of high transmission areas $(R_0 > 10)$ and mapping assumptions as for the upper map. Areas that appear to be uniform may have small-scale heterogeneities in transmission that are beyond the scale of this map.

(PDF)

Figure S5 Maps of the predicted probabilities that a ten-fold effect size will interrupt malaria transmission. The upper and lower maps are illustrated as per **Figure S4**, except that **Figure S5** shows the predicted probabilities that a ten-fold reduction in transmission ('ten-fold effect size') would interrupt malaria transmission over a given pixel (size is 5 km²). (PDF)

14

Text S1 Pharmacokinetic and pharmacodynamic equations and distributions [71–90]. (PDF)

Acknowledgments

Computing resources were provided in part by the 'Hotfoot' High Performance Computing Cluster at Columbia University.

References

- 1. WHO (2012) World malaria report: 2012. Switzerland: WHO Press.
- Hay SI, Okiro EA, Gething PW, Patil AP, Tatem AJ, et al. (2010) Estimating the global clinical burden of *Plasmodium falciparum* malaria in 2007. PLoS Med 7: e1000290.
- Murray CJL, Rosenfeld LC, Lim SS, Andrews KG, Foreman KJ, et al. (2012) Global malaria mortality between 1980 and 2010: a systematic analysis. Lancet 379: 413–431.
- Eastman RT, Fidock DA (2009) Artemisinin-based combination therapies: a vital tool in efforts to eliminate malaria. Nat Rev Microbiol 7: 864–874.
- White NJ (2008) Qinghaosu (artemisinin): the price of success. Science 320: 330– 334.
- Smith DL, Cohen JM, Moonen B, Tatem AJ, Sabot OJ, et al. (2011) Infectious disease. Solving the Sisyphean problem of malaria in Zanzibar. Science 332: 1384–1385.
- Chiyaka C, Tatem A, Cohen J (2013) The stability of malaria elimination. Science 339: 909–910.
- Feachem R, Sabot O (2008) A new global malaria eradication strategy. Lancet 371: 1633–1635.
- Feachem RG, Phillips AA, Targett GA, Snow RW (2010) Call to action: priorities for malaria elimination. Lancet 376: 1517–1521.
- Dondorp AM, Fairhurst RM, Slutsker L, Macarthur JR, M DJ, et al. (2011) The threat of artemisinin-resistant malaria. N Engl J Med 365: 1073–1075.
- Trape JF, Tall A, Diagne N, Ndiath O, Ly AB, et al. (2011) Malaria morbidity and pyrethroid resistance after the introduction of insecticide-treated bednets and artemisinin-based combination therapies: a longitudinal study. Lancet Infect Dis 11: 925–932.
- Paul RE, Hackford I, Brockman A, Muller-Graf C, Price R, et al. (1998) Transmission intensity and *Plasmodium falciparum* diversity on the northwestern border of Thailand. Am J Trop Med Hyg 58: 195–203.
- Pigott DM, Atun R, Moyes CL, Hay SI, Gething PW (2012) Funding for malaria control 2006–2010: A comprehensive global assessment. Malaria J 11: 246.
- Greenwood BM, Fidock DA, Kyle DE, Kappe SH, Alonso PL, et al. (2008) Malaria: progress, perils, and prospects for eradication. The Journal of clinical investigation 118: 1266–1276.
- Cohen JM, Moonen B, Snow RW, Smith DL (2010) How absolute is zero? An evaluation of historical and current definitions of malaria elimination. Malaria J 9: 213.
- Sabot O, Cohen JML, Hsiang MS, Kahn JG, Basu S, et al. (2010) Costs and financial feasibility of malaria elimination. Lancet 376: 1604–1615.
- Tatem AJ, Smith DL, Gething PW, Kabaria CW, Snow RW, et al. (2010) Ranking of elimination feasibility between malaria-endemic countries. Lancet 376: 1579–1591.
- Anonymous (2011) The MalERA Consultative Group on Modeling. A research agenda for malaria eradication: modeling. PLoS Med 8: e1000403.
- White NJ (2008) The role of anti-malarial drugs in eliminating malaria. Malaria J 7 Suppl 1: S8.
- Vale N, Moreira R, Gomes P (2009) Primaquine revisited six decades after its discovery. Eur J Med Chem 44: 937–953.
- Adjalley SH, Johnston GL, Li T, Eastman RT, Ekland EH, et al. (2011) Quantitative assessment of *Plasmodium falciparum* sexual development reveals potent transmission-blocking activity by methylene blue. Proc Natl Acad Sci USA 108: E1214–1223.
- 22. Muller O, Mockenhaupt FP, Marks B, Meissner P, Coulibaly B, et al. (2013) Haemolysis risk in methylene blue treatment of G6PD-sufficient and G6PDdeficient West-African children with uncomplicated falciparum malaria: a synopsis of four RCTs. Pharmacoepidemiology and drug safety 22: 376–385.
- Johnston GL, Smith DL, Fidock DA (2013) Malaria's missing number: calculating the human component of R0 by a within-host mechanistic model of *Plasmodium falciparum* infection and transmission. PLoS Comput Biol 9: e1003025.
- Yeung S, Van Damme W, Socheat DD, White NJ, Mills A (2008) Access to artemisinin combination therapy for malaria in remote areas of Cambodia. Malaria J 7: 96.
- Wells TN, Alonso PL, Gutteridge WE (2009) New medicines to improve control and contribute to the eradication of malaria. Nature reviews Drug discovery 8: 879–891.
- Bousema T, Drakeley C (2011) Epidemiology and infectivity of *Plasmodium falciparum* and *Plasmodium vivax* gametocytes in relation to malaria control and elimination. Clinical microbiology reviews 24: 377–410.

Author Contributions

Conceived and designed the experiments: GLJ PWG SIH DLS DAF. Performed the experiments: GLJ PWG SIH. Analyzed the data: GLJ PWG SIH DLS DAF. Wrote the paper: GLJ DLS DAF.

- Bousema T, Okell LC, Shekalaghe S, Griffin JT, Omar SA, et al. (2010) Revisiting the circulation time of *Plasmodium falciparum* gametocytes: molecular detection methods to estimate the duration of gametocyte carriage and the effect of gametocytocidal drugs. Malaria J 9: 136.
- Bousema JT, Schneider P, Gouagna LC, Drakeley CJ, Tostmann A, et al. (2006) Moderate effect of artemisinin-based combination therapy on transmission of *Plasmodium falciparum*. J Infect Dis 193: 1151–1159.
- Molineaux L, Diebner HH, Eichner M, Collins WE, Jeffery GM, et al. (2001) *Plasmodium falciparum* parasitaemia described by a new mathematical model. Parasitology 122: 379–391.
- Targett G, Drakeley C, Jawara M, von Seidlein L, Coleman R, et al. (2001) Artesunate reduces but does not prevent posttreatment transmission of *Plasmodium falciparum* to *Anopheles gambiae*. J Infect Dis 183: 1254–1259.
- Drakeley ČJ, Jawara M, Targett GaT, Walraven G, Obisike U, et al. (2004) Addition of artesunate to chloroquine for treatment of *Plasmodium falciparum* malaria in Gambian children causes a significant but short-lived reduction in infectiousness for mosquitoes. Trop Med Int Health 9: 53–61.
- Sutherland CJ, Ord R, Dunyo S, Jawara M, Drakeley CJ, et al. (2005) Reduction of malaria transmission to *Anopheles* mosquitoes with a six-dose regimen of co-artemether. PLoS Med 2: e92.
- Hallett R, Dunyo S, Ord R, Jawara M, Pinder M, et al. (2006) Chloroquine/ sulphadoxine-pyrimethamine for Gambian children with malaria: transmission to mosquitoes of multidrug-resistant *Plasmodium falciparum*. PLoS Clin Trials 1: E14.
- Piyaphanee W, Krudsood S, Tangpukdee N, Thanachartwet W, Silachamroon U, et al. (2006) Emergence and clearance of gametocytes in uncomplicated *Plasmodium falciparum* malaria. Am J Trop Med Hyg 74: 432–435.
- Shekalaghe S, Drakeley C, Gosling R, Ndaro A, van Meegeren M, et al. (2007) Primaquine clears submicroscopic *Plasmodium falciparum* gametocytes that persist after treatment with sulphadoxine-pyrimethamine and artesunate. PLoS ONE 2: 1–8.
- Smithuis F, Kyaw MK, Phe O, Win T, Aung PP, et al. (2010) Effectiveness of five artemisinin combination regimens with or without primaquine in uncomplicated falciparum malaria: an open-label randomised trial. Lancet Infect Dis 10: 673–681.
- Sawa P, Shekalaghe Sa, Drakeley CJ, Sutherland CJ, Mweresa CK, et al. (2013) Malaria transmission after artemether-lumefantrine and dihydroartemisininpiperaquine: a randomized trial. J Infect Dis 10.1093/infdis/jit077: 1–9.
- 38. Sutanto I, Suprijanto S, Kosasih A, Dahlan MS, Syafruddin D, et al. (2013) The effect of primaquine on gametocyte development and clearance in the treatment of uncomplicated falciparum malaria with dihydroartemisinin-piperaquine in South sumatra, Western indonesia: an open-label, randomized, controlled trial. Clin Infect Dis 56: 685–693.
- Lawpoolsri S, Klein EY, Singhasivanon P, Yimsamran S, Thanyavanich N, et al. (2009) Optimally timing primaquine treatment to reduce *Plasmodium falciparum* transmission in low endemicity Thai-Myanmar border populations. Malaria J 8: 159.
- Smith T, Ross A, Maire N, Chitnis N, Studer A, et al. (2012) Ensemble modeling of the likely public health impact of a pre-erythrocytic malaria vaccine. PLoS Med 9: e1001157.
- Gurarie D, Karl S, Zimmerman Pa, King CH, St Pierre TG, et al. (2012) Mathematical modeling of malaria infection with innate and adaptive immunity in individuals and agent-based communities. PLoS ONE 7: e34040.
- Okell LC, Ghani AC, Lyons E, Drakeley CJ (2009) Submicroscopic infection in *Plasmodium falciparum*—endemic populations: a systematic review and metaanalysis. J Infect Dis 200: 1509–1517.
- 43. White NJ, Qiao LG, Qi G, Luzzatto L (2012) Rationale for recommending a lower dose of primaquine as a *Plasmodium falciparum* gametocytocide in populations where G6PD deficiency is common. Malaria J 11: 418.
- White NJ (2013) Primaquine to prevent transmission of falciparum malaria. Lancet Infect Dis 13: 175–181.
- Gething PW, Patil AP, Smith DL, Guerra CA, Elyazar IR, et al. (2011) A new world malaria map: *Plasmodium falciparum* endemicity in 2010. Malaria J 10: 378.
- Smith DL, Hay SI (2009) Endemicity response timelines for *Plasmodium falciparum* elimination. Malaria J 8: 87.
- Smith DL, McKenzie FE, Snow RW, Hay SI (2007) Revisiting the basic reproductive number for malaria and its implications for malaria control. PLoS biology 5: e42.
- Luxemburger C, Ricci F, Nosten F, Raimond D, Bathet S, et al. (1997) The epidemiology of severe malaria in an area of low transmission in Thailand. Trans R Soc Trop Med Hyg 91: 256–262.

- Griffin JT, Hollingsworth TD, Okell LC, Churcher TS, White M, et al. (2010) Reducing *Plasmodium falciparum* malaria transmission in Africa: a model-based evaluation of intervention strategies. PLoS Med 7: e1000324.
- Maude RJ, Socheat D, Nguon C, Saroth P, Dara P, et al. (2012) Optimising strategies for *Plasmodium falciparum* malaria elimination in Cambodia: primaquine, mass drug administration and artemisinin resistance. PLoS One 7: e37166.
- Phyo AP, Nkhoma S, Stepniewska K, Ashley EA, Nair S, et al. (2012) Emergence of artemisinin-resistant malaria on the western border of Thailand: a longitudinal study. Lancet 379: 1960–1966.
- 53. Fidock DA (2013) Malaria elimination. Science 340: 1531-1533.
- Diebner HH, Eichner M, Molineaux L, Collins WE, Jeffery GM, et al. (2000) Modelling the transition of asexual blood stages of *Plasmodium falciparum* to gametocytes. J Theor Biol 202: 113–127.
- Eichner M, Diebner HH, Molineaux L, Collins WE, Jeffery GM, et al. (2001) Genesis, sequestration and survival of *Plasmodium falciparum* gametocytes: parameter estimates from fitting a model to malariatherapy data. Trans R Soc Trop Med Hyg 95: 497–501.
- 56. Smith TA, Killeen GF, Maire N, Ross A, Molineaux L, et al. (2006) Mathematical modeling of the impact of malaria vaccines on the clinical epidemiology and natural history of *Plasmodium falciparum* malaria: overview. Am J Trop Med Hyg 75: 1–10.
- Crooks L (2008) Problems with continuous-time malaria models in describing gametocytogenesis. Parasitology 135: 881–896.
- Kay K, Hastings IM (2013) Improving pharmacokinetic-pharmacodynamic modeling to investigate anti-infective chemotherapy with application to the current generation of antimalarial drugs. PLoS Comput Biol 9: e1003151.
- Winter K, Hastings IM (2011) Development, evaluation, and application of an in silico model for antimalarial drug treatment and failure. Antimicrob Agents Chemother 55: 3380–3392.
- Karunajeewa HA, Ilett KF, Mueller I, Siba P, Law I, et al. (2008) Pharmacokinetics and efficacy of piperaquine and chloroquine in Melanesian children with uncomplicated malaria. Antimicrob Agents Chemother 52: 237– 243.
- Ezzet F, van Vugt M, Nosten F, Looareesuwan S, White NJ (2000) Pharmacokinetics and pharmacodynamics of lumefantrine (benflumetol) in acute falciparum malaria. Antimicrob Agents Chemother 44: 697–704.
- 62. Krudsood S, Looareesuwan S, Tangpukdee N, Wilairatana P, Phumratanaprapin W, et al. (2010) New fixed-dose artesunate-mefloquine formulation against multidrug-resistant *Plasmodium falciparum* in adults: a comparative phase IIb safety and pharmacokinetic study with standard-dose nonfixed artesunate plus mefloquine. Antimicrob Agents Chemother 54: 3730–3737.
- 63. Gutman J, Green M, Durand S, Rojas OV, Ganguly B, et al. (2009) Mefloquine pharmacokinetics and mefloquine-artesunate effectiveness in Peruvian patients with uncomplicated *Plasmodium falciparum* malaria. Malaria J 8: 58.
- Prinz H (2010) Hill coefficients, dose-response curves and allosteric mechanisms. J Chem Biol 3: 37–44.
- Simpson JAA, Watkins ERR, Price RNN, Aarons L, Kyle DEE, et al. (2000) Mefloquine pharmacokinetic-pharmacodynamic models: implications for dosing and resistance. Antimicrob Agents Chemother 44: 3414.
- Hoshen MB, Stein WD, Ginsburg HD (2001) Pharmacokinetic-pharmacodynamic modelling of the antimalarial activity of mefloquine. Parasitology 123: 337–346.
- Bousema T, Griffin JT, Sauerwein RW, Smith DL, Churcher TS, et al. (2012) Hitting hotspots: spatial targeting of malaria for control and elimination. PLoS Med 9: e1001165.
- Beavogui A, Djimde A, Gregson A (2010) Low infectivity of *Plasmodium falciparum* gametocytes to *Anopheles gambiae* following treatment with sulfadoxine–pyrimethamine in Mali. Int J Parasitol 40: 1213–1220.
- Kone A, van de Vegte-Bolmer M, Siebelink-Stoter R, van Gemert G-J, Dara A, et al. (2010) Sulfadoxine-pyrimethamine impairs *Plasmodium falciparum* gametocyte infectivity and *Anopheles* mosquito survival. Int J Parasitol 40: 1221–1228.
- Delves M, Plouffe D, Scheurer C, Meister S, Wittlin S, et al. (2012) The activities of current antimalarial drugs on the life cycle stages of *Plasmodium*: a comparative study with human and rodent parasites. PLoS Med 9: e1001169.

- Ezzet F, Mull R, Karbwang J (1998) Population pharmacokinetics and therapeutic response of CGP 56697 (artemether+benflumetol) in malaria patients. Brit J Clin Pharmacol 46: 553–561.
- White NJ, van Vugt M, Ezzet F (1999) Clinical pharmacokinetics and pharmacodynamics and pharmacodynamics of artemether-lumefantrine. Clin Pharmacokinet 37: 105–125.
- Newton P, Suputtamongkol Y, Teja-Isavadharm P, Pukrittayakamee S, Navaratnam V, et al. (2000) Antimalarial bioavailability and disposition of artesunate in acute falciparum malaria. Antimicrob Agents Chemother 44: 972– 977.
- 74. van Vugt M, Brockman A, Gemperli B, Luxemburger C, Gathmann I, et al. (1998) Randomized comparison of artemether-benflumetol and artesunatemefloquine in treatment of multidrug-resistant falciparum malaria. Antimicrob Agents Chemother 42: 135–139.
- Lefevre G, Looareesuwan S, Treeprasertsuk S, Krudsood S, Silachamroon U, et al. (2001) A clinical and pharmacokinetic trial of six doses of artemetherlumefantrine for multidrug-resistant *Plasmodium falciparum* malaria in Thailand. Am J Trop Med Hyg 64: 247–256.
- Lefevre G, Bindschedler M, Ezzet F, Schaeffer N, Meyer I, et al. (2000) Pharmacokinetic interaction trial between co-artemether and mefloquine. Eur J Pharm Sci 10: 141–151.
- Ashley EA, Stepniewska K, Lindegardh N, McGready R, Hutagalung R, et al. (2006) Population pharmacokinetic assessment of a new regimen of mefloquine used in combination treatment of uncomplicated falciparum malaria. Antimicrob Agents Chemother 50: 2281–2285.
- 78. Thummers B (2006) DataThiefIII.
- Bertrand J, Mentre F (2008) Mathematical expressions of the pharmacokinetic and pharmacodynamic models implemented in the Monolix software. Available at: http://wwwfacmuclacbe/cooperation/Vietnam/WBI-Vietnam-October-2011/Modelling/Monolix32_PKPD_librarypdf.
- Price RN, Uhlemann AC, van Vugt M, Brockman A, Hutagalung R, et al. (2006) Molecular and pharmacological determinants of the therapeutic response to artemether-lumefantrine in multidrug-resistant *Plasmodium falciparum* malaria. Clin Infect Dis 42: 1570–1577.
- Hoshen MB, Stein WD, Ginsburg H (1998) Modelling the chloroquine chemotherapy of falciparum malaria: the value of spacing a split dose. Parasitology 116 (Pt 5): 407–416.
- Sidhu AB, Verdier-Pinard D, Fidock DA (2002) Chloroquine resistance in *Plasmodium falciparum* malaria parasites conferred by *pfcrt* mutations. Science 298: 210–213.
- Bustos MD, Gay F, Diquet B (1994) In vitro tests on Philippine isolates of Plasmodium falciparum against four standard antimalarials and four qinghaosu derivatives. Bull WHO 72: 729–735.
- van Agtmael MA, Cheng-Qi S, Qing JX, Mull R, van Boxtel CJ (1999) Multiple dose pharmacokinetics of artemether in Chinese patients with uncomplicated falciparum malaria. Int J Antimicrob Agents 12: 151–158.
- Cousin M, Kummerer S, Lefevre G, Marrast AC, Stein D, et al. (2008) Antiinfective drugs advisory committee meeting: Coartem (artemether-lumefantrine) tablets for the treatment of malaria in patients with acute, uncomplicated infections due to *Plasmodium falciparum* or mixed infections including *P. falciparum*. Available at: http://wwwfdagov/ohrms/dockets/ac/08/briefing/2008-4388b1-02-Novartispdf [Accessed November 1, 2012]: 117.
- ter Kuile FO, Nosten F, Thieren M, Luxemburger C, Edstein MD, et al. (1992) High-dose mefloquine in the treatment of multidrug-resistant falciparum malaria. J Infect Dis 166: 1393–1400.
- White NJ (1997) Assessment of the pharmacodynamic properties of antimalarial drugs in vivo. Antimicrob Agents Chemother 41: 1413–1422.
- Okell LC, Drakeley CJ, Ghani AC, Bousema T, Sutherland CJ (2008) Reduction of transmission from malaria patients by artemisinin combination therapies: a pooled analysis of six randomized trials. Malaria J 7: 125.
- Hallett RL, Sutherland CJ, Alexander N, Ord R, Jawara M, et al. (2004) Combination therapy counteracts the enhanced transmission of drug-resistant malaria parasites to mosquitoes. Antimicrob Agents Chemother 48: 3940–3943.
- van Pelt-Koops JC, Pett HE, Graumans W, van der Vegte-Bolmer M, van Gemert GJ, et al. (2012) The spiroindolone drug candidate NITD609 potently inhibits gametocytogenesis and blocks *Plasmodium falciparum* transmission to anopheles mosquito vector. Antimicrob Agents Chemother 56: 3544–3548.